Author's response to reviews

Title: Obstetrician-Gynecologists’ Knowledge of Sickle Cell Disease Screening and Management

Authors:

Ijeoma Azonobi (iazonobi@msm.edu)  
Britta L Anderson (banderson@acog.org)  
Vanessa Byams (ver0@cdc.gov)  
Althea Grant (ang4@cdc.gov)  
Jay Schulkin (jschulkin@acog.org)

Version: 2  
Date: 16 April 2014

Author’s response to reviews:

Dear Editor, We have made the requested formatting changes. Please let me us know if there is anything else we can do at this time.

Sincerely,  
The authors